

Citation: Skov L, Hui R, Shchur V, Hobolth A, Scally A, Schierup MH, et al. (2018) Detecting archaic introgression using an unadmixed outgroup. PLoS Genet 14(9): e1007641. https:// doi.org/10.1371/journal.pgen.1007641

Editor: Fernando Racimo, University of Copenhagen, DENMARK

Received: March 27, 2018

Accepted: August 17, 2018

Published: September 18, 2018

Copyright: © 2018 Skov et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: This study uses third party genomic data from Papuan individuals. Whole-genome sequence data are available from dbGAP with the accession number phs001085.v1. p1 (Vernot 2016) and interested researchers would need to apply for access by contacting Joshua Akey (akeyj@uw.edu). The Aboriginal Australian and Papuan whole genome sequence data are available from European Genome-phenome Archive (EGA, http://www.ebi.ac.uk/ega/), which is hosted by the EBI, under the accession numbers EGAS00001001766 and EGAS00001001247, respectively (Malaspinas 2016). Raw data for 279 **RESEARCH ARTICLE**

Detecting archaic introgression using an unadmixed outgroup

Laurits Skov^{1*}, Ruoyun Hui², Vladimir Shchur³, Asger Hobolth¹, Aylwyn Scally², Mikkel Heide Schierup¹, Richard Durbin^{2,3*}

1 Bioinformatics Research Centre, Aarhus University, Aarhus C., Denmark, 2 Department of Genetics, University of Cambridge, Cambridge United Kingdom, 3 Wellcome Sanger Institute, Hinxton, Cambridge, United Kingdom

* lskov@cs.au.dk(LS); rd@sanger.ac.uk(RD)

Abstract

Human populations outside of Africa have experienced at least two bouts of introgression from archaic humans, from Neanderthals and Denisovans. In Papuans there is prior evidence of both these introgressions. Here we present a new approach to detect segments of individual genomes of archaic origin without using an archaic reference genome. The approach is based on a hidden Markov model that identifies genomic regions with a high density of single nucleotide variants (SNVs) not seen in unadmixed populations. We show using simulations that this provides a powerful approach to identifying segments of archaic introgression with a low rate of false detection, given data from a suitable outgroup population is available, without the archaic introgression but containing a majority of the variation that arose since initial separation from the archaic lineage. Furthermore our approach is able to infer admixture proportions and the times both of admixture and of initial divergence between the human and archaic populations. We apply the model to detect archaic introgression in 89 Papuans and show how the identified segments can be assigned to likely Neanderthal or Denisovan origin. We report more Denisovan admixture than previous studies and find a shift in size distribution of fragments of Neanderthal and Denisovan origin that is compatible with a difference in admixture time. Furthermore, we identify small amounts of Denisova ancestry in South East Asians and South Asians.

Author summary

The genetic history of present-day individuals includes episodes of mating between divergent groups, which have led to 'introgressed' genetic material persisting in modern genome sequences. Perhaps the most notable examples of such events in humans are the introgressions from Neanderthals into non-Africans 50,000 or so years ago, and from a related archaic group known as Denisovans into the ancestors of indigenous people from Papua-New Guinea and Australia. Methods to identify introgressions and the genomic regions that derive from them generally involve the use of reference genome sequences for the source populations. However, there are advantages in having methods independent of reference sequences, both to reduce bias and to detect possible introgression from genomes for which the informed consent documentation is consistent with fully public data release are available through the EBI European Nucleotide Archive under accession numbers PRJEB9586 and ERP010710 (Simons Genome Diversity Project).

Funding: LS was supported by grants 1323-00076 and 6108-00385 from the Danish Council for Independent Research, Natural Sciences (to MHS). https://ufm.dk/en/research-and-innovation/ councils-and-commissions/independent-researchfund-Denmark RH was supported by a Gates Cambridge Scholarship. RD was supported by Wellcome Trust grants WT206194 and WT207492. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. https://wellcome.ac. uk/funding

Competing interests: The authors have declared that no competing interests exist.

groups for which we currently lack a reference genome. In this paper we describe such an approach, in a statistical framework which exploits the fact that introgressed regions will contain a high density of genetic variants that are private to the group receiving the divergent material. We apply this method to 89 Papuan genome sequences, estimating times of introgression and initial divergence between archaic and modern humans, and compare it to other related methods.

Introduction

Archaic introgression into modern humans occurred at least twice (Neanderthals and Denisovans) [1,2] and had a phenotypic effect on humans [3–5]. A substantial amount of Neanderthal and Denisovan genetic material is still present in modern humans and we can learn about archaic populations from studying their genetic variants in humans.

To harness this information a number of methods have been developed to infer segments of archaic ancestry in an individual's genome. Scanning along the genome, Hidden Markov Models (HMMs)[1,6] and Conditional Random Fields (CRFs)[7] can identify haplotype segments in non-Africans that are both closer to the archaic reference genomes than to Africans, and also longer than expected by incomplete lineage sorting; these are then identified as likely archaic introgressed segments. Another approach is to identify segments with more variants in high linkage disequilibrium (LD) that are unique to non-Africans than expected given a certain demographic scenario [8]. The latest implementations of this method also use an archaic reference genome for refining the set of putative archaic haplotypes [9].

The use of archaic reference genomes for identification of introgressed fragments has drawbacks. First, since the introgressing Neanderthal is closer to the Neanderthal reference genomes (80,000–145,000 years divergence)[10], than the introgressing Denisova is to the Denisova genome (276,000–403,000 years divergence)[1] detecting Denisovan ancestry will be harder. Second, the reliance on having reference genomes implies that the introgression maps need updates whenever more archaic reference genomes are sequenced [10]. Finally, it may be hard to identify potential introgressed segments from an unknown archaic origin, as in the case of the putative archaic introgression into Pygmies [11] and Andamanese islanders [12]. There have been previous approaches that do not use reference genomes, including the initial version of S^{*} [8] and recent methods [13,14].

Here we present a new method for the identification of archaic segments that does not require an archaic reference genome but does require an outgroup population that does not contain admixed genetic material from the archaic population. We implement an HMM that examines the density of variants unseen in the outgroup along individual genomes. Because our method is based on a demographic model incorporating mutation and recombination over time, we can estimate demographic parameters relevant to introgression. We demonstrate with Papuans how we can estimate such parameters and infer more archaic material than previously. Furthermore we can separate the archaic material into Denisovan and Neanderthal components, which display different length distributions in accordance with different admixture times.

Overview of the model

An archaic genomic segment introgressed into a population is expected to have a high density of variants not found in populations without the introgression. We use a Hidden Markov Model (HMM) to classify genomic segments into states with varying density of such variants.

We focus on a scenario where introgression from a deeply divergent archaic population only happened into an ingroup and not the outgroup, see Fig 1A. By removing variants found in the outgroup we can better distinguish introgressed segments from non-introgressed segments based on the density of remaining variants, see Fig 1A. These remaining variants, which we denote private variants (because they are private to the ingroup with respect to the outgroup) can either have occurred on the branch starting from the split of the ingroup and outgroup, or on the introgressing population's branch. Because the introgressed segments have had a longer time to accumulate variants, they should have a higher density of private variants.

Thus, we define a HMM with two states. The hidden states are Ingroup and Archaic, and the probability for changing state in the Ingroup is p and the probability for changing state in the Archaic is q, see Fig 1B. The probability of changing state can also be expressed in terms of a constant recombination rate between windows $r \cdot L$, the admixture time T_{admix} and admixture proportion a, see Fig 1B.



Fig 1. Overview of the model. Illustration on small test dataset. a) An archaic segment introgresses into the ingroup population at time T_{achnix} with admixture proportion *a*. The segments in the ingroup have a mean coalescence time with a segment from the outgroup at time $T_{Ingroup}$ and an archaic segment has a mean coalescence time with a segment from the outgroup at time $T_{Archaic}$. Removing all variants found in the outgroup (light orange points) should remove all the variants in the common ancestor of ingroup and outgroup, leaving only private variants that either occurred on the ingroup branch (dark orange) or on the archaic branch (dark blue). This will make the archaic segment have a higher variant density. The genome is then binned into windows of length *L* (here 1000 bp) and the number of private variants is counted in each window. These are the observations and the hidden states are either Ingroup state or Archaic state. When decoding the sequence the most likely path through the sequence is found. b) The transition matrix between the archaic state and ingroup state. c) The emission probabilities are modelled as Poisson distributions with means $\lambda_{Ingroup}$ and $\lambda_{Archaic}$. It is more likely to see more private variants in the Archaic state than in the Ingroup state.

https://doi.org/10.1371/journal.pgen.1007641.g001

For practical purposes we bin the genome into windows of length L (typically L = 1000 bp). The number of private variants observed in a window is Poisson distributed with a rate $\lambda_{Ingroup}$ and $\lambda_{Archaic}$, respectively where $\lambda_{Ingroup} = \mu \cdot L \cdot \lambda_{Ingroup}$ and $\lambda_{Archaic} = \mu \cdot L \cdot \lambda_{Archaic}$, μ is the mutation rate, $T_{Ingroup}$ is the mean coalescence time for the ingroup and the outgroup and $T_{archaic}$ is the mean coalescence time for the archaic population and the outgroup, see Fig 1C.

We make a correction to the rates to take into account the number of missing bases in a window and the local mutation rate. For window *i* we have $\lambda_{Ingroup}^{i} = \mu_{i} \cdot L_{i} \cdot T_{Ingroup}$ and $\lambda_{Archaic}^{i} = \mu_{i} \cdot L_{i} \cdot T_{Archaic}$, where μ_{i} is the local mutation rate and L_{i} is the number of called bases in a window.

The set of transition parameters p, q and the Poisson parameters $\lambda_{Ingroup}$, $\lambda_{Archaic}$ that maximize the likelihood given the observations are found using the Baum-Welch algorithm for an individual genome. These parameters are informative of the mean coalescence times between the ingroup and outgroup and between the archaic and the outgroup, the admixture time and the admixture proportion if we assume a known mutation rate μ and a known recombination rate between windows *rL*. Once the set of optimal parameters are found they can be used to decode the genome, using posterior decoding to identify candidate introgressed segments as consecutive regions with posterior probability of coming from the archaic state above some threshold.

Until now we have assumed the data is phased haploid genomes. But to avoid problems with phasing we run this model on unphased diploid genomes. Heterozygous archaic segments will still stand out from homozygous non-introgressed segments. Formally this is equivalent to assuming that homozygous introgressed segments are sufficiently rare that they can be ignored for model fitting. In practice any homozygous archaic segments will have higher private variant density than heterozygous segments, so in the absence of a homozygous HMM state they will be classified with the heterozygous state. We show how to convert model parameters to demographic parameters, both when analyzing haploid and diploid genomes in S1 Dataset.

We note that this method will likely only work in cases where the coalescence time distribution of the ingroup and archaic segments are sufficiently different. This will work better in cases where the variation in the ingroup is a subset of variation in the outgroup so the majority of variation in the common ancestor can be removed, as the case of Non-Africans and Africans.

Results

Testing the model with simulations

To investigate the ability of our model to identify archaic (Neanderthal and Denisovan) admixture into Papuans we simulated whole diploid autosomal data using a coalescent simulator, with admixture with an archaic hominin 1,500 generations ago replacing 0-25% of the population–(a script with all demographic parameters is shown in S2 Dataset and a graphical representation of the demography is shown in S1 Fig). We simulated different scenarios to test the effects of running the model on haploid versus diploid data, adding missing data, varying recombination rate and varying a mutation rate.

First, we simulated five individuals where every base in the genome is called equally well and there is a constant recombination rate of $1.2 \cdot 10^{-8}$ events per basepair per generation. We call this dataset the ideal data. Second, we simulated five individuals with missing data (using the repeatmask track for the human reference genome hg19 [15]) and variations in local recombination rate (using HapMap phase II [16]) to test the effect of missing data and recombination. Third we add variations in local mutation rate to the second scenario as described in the materials and method section. We binned all genomes into bins of 1000 bp, and removed all variants found in any of 500 simulated Africans, 100 simulated Europeans and 100 simulated Asians. We train the model on both haplotype data and unphased diploid genotype data for the simulated individuals. The latter is similar to situations where phased data is not available.

We estimated the transition and emission parameters using the Baum-Welch algorithm and used them to get an estimate for the admixture time T_{admix} , the admixture proportion *a* and the mean coalescent times with the outgroup $T_{Ingroup}$ and $T_{Archaic}$ for the ingroup and archaic segments respectively. We also show the sensitivity and precision of the model at different admixture proportions. We only show the estimated parameters and error rates for simulations with missing data and varying recombination rate, because the addition of a varying mutation rate has a very minor effect, see Fig 2B. A table containing all parameters from the model and the corresponding demographic parameters are listed in S3 Dataset.



Fig 2. Evaluation of the model on simulated data. a). The estimated parameters T_{admix} , $a, T_{Ingroup}$ and $T_{Archaic}$ are shown for different admixture proportions in simulated data with varying recombination rate and missing data. We also show the sensitivity and precision for different admixture proportions. For sensitivity and precision we show the values with a posterior probability cutoff at 0.5 (average posterior probability of all bins being belonging to the archaic state for a segments) b). Sensitivity and precision shown for the Sstar methods, Sprime and the HMM on different datasets. For Sstar and Sprime methods the different points are when the score for a segment is 50,000, 100,000, 150,000 and 200,000 as in Browning et al 2018. For the HMM the cutoffs is 0.5, 0.6, 0.7, 0.8 and 0.9. c) When there is no admixture the model is not in agreement with itself. The estimated admixture proportion from the transition matrix does not match the amount of sequence classified as belonging to the archaic state.

https://doi.org/10.1371/journal.pgen.1007641.g002

We evaluate the performance of the model in terms of precision (amount of predicted archaic sequence that is archaic/amount of predicted archaic sequence) and sensitivity (amount of predicted archaic sequence).

When admixture proportions are low (less than 2%) the model does not fit the emission or transition parameters well and the precision is around 70%. With an admixture proportion greater than 2% the model fits the parameters well with sensitivity above 80% and precision greater than 80% at a posterior probability cutoff at 0.5 (mean posterior probability of being archaic for all windows in segment). Raising the posterior probability cutoff from 0.5 to 0.8 increases the precision to around 90% while the sensitivity is still above 75% for 5% admixture as can be seen in S3 Fig. The effect of changing the cutoff for different admixture proportions are shown in S3 Dataset.

Across all scenarios $T_{Ingroup}$ is slightly underestimated for data sets, while $T_{Archaic}$ is overestimated, see Fig 2A. The error in estimating $T_{Ingroup}$ and $T_{Archaic}$ is likely due to model misspecification: the model effectively represents the distribution of coalescence times with the mean coalescence time only, and assumes all sites are heterozygous for both states. The effect of treating all sites as heterozygous is seen when comparing the $T_{Ingroup}$ for haploid (mean = 2,952 generations ago) and diploid data (mean = 2,646 generations ago). The average simulated $T_{Ingroup}$ was 3,109 generations ago. The effect is also seen when comparing $T_{Archaic}$ for haploid (mean = 36,844 generations ago) and diploid data (mean = 38,508 generations ago). The average simulated $T_{Archaic}$ was 36,462 generations ago.

Furthermore, the model tends to classify deeply coalescing haplotypes from the common ancestor of ingroup and outgroup as admixed. This effect is greater in simulations with low admixture proportions (Fig 2A).

The misspecification of the coalescent times is more problematic in cases where the ancestral population of ingroup and outgroup is very large and/or contains strong population structure. This can be overcome to some extent by sequencing more individuals from the outgroup, but the improvement becomes limited if there have been bottlenecks in the outgroup. The effective population size of the archaic source population after its separation from the common ancestor of ingroup and outgroup does not matter, since we assume very few lineages contribute to the admixture in this population that affects the test sample. Finally, a large population size in the ancestral population will increase the variance in coalescent times in the archaic state, but we would expect this to have less consequence on the models ability to discriminate.

We find that when admixture proportion are >5% and the recombination rate across the genome is constant the model recovers the right admixture time (<u>S3 Dataset</u>). However with varying recombination rate we underestimate the admixture time, which might be due to the model's failure to identify around 80% of the short segments (see <u>S2 Fig</u>). This would increase the average segment length and lead to a more recent admixture time.

The admixture proportion is fitted well for all simulations except where the admixture proportion is zero. In this case, the "archaic" state is assigned to a set of segments with longer coalescent times to lineages in the outgroup, but their posterior probability is lower than for real admixed segments. The inconsistency between the estimated admixture proportion and the amount of segments recovered by posterior decoding, could potentially be used to discriminate whether or not there is admixture (Fig 2C). For admixture proportions below 2% (0.5% for data with varying recombination rate) we observe that T_{admix} is estimated to be greater than $T_{Ingroup}$ (meaning admixture happened before the split of ingroup and outgroup), which is not possible and also indicates breakdown of the model.

We also compared the performance of our method to Sstar 2014[17], Sstar 2016[9] and Sprime 2018[13] under scenarios of varying recombination rate, varying recombination rate and missing data and varying mutation rate, missing data and varying mutation rate. Our

method shows improved tradeoff between sensitivity and precision, because we take missing data into account when training and decoding the model, (Fig 2B and S3 Fig).

Application to Papuan genomes

Having validated the model, we applied it to 14 Papuan individuals from the Simons Genome Diversity Project [18], 40 Papuans from [19] and an additional 35 Papuans [9]. We also analyzed individuals from West Eurasia, East Asia and South East Asia from the Simons Genome Diversity Project[18].

We note that variants in these datasets have been found using different bioinformatics pipelines with different filters but that the counts of heterozygous and homozygous variants are similar, see <u>S4 Dataset</u>.

We estimate the background mutation rate in windows of 100 kb, using the variant density of all variants in African populations from the 1000 Genomes Project.

Our model will not be able to distinguish Neanderthal from Denisova segments in Papuans, because the Denisovans and Neanderthals share a common ancestor before they do with humans and therefore the mean coalescence time with humans will be the same [1]. This means that the Poisson parameters will be the same as they both depend on $T_{Archaic}$. However, we are able to enrich for Denisova versus Neanderthal segments by using different outgroups in our filtering step, because Neanderthal ancestry is common to all non-African populations whereas Denisovan ancestry relatively more private to Melanesia [7,9].

For each individual we used two different sets of variants as outgroup.

First, we used only variants found in Sub-Saharan African populations as an outgroup (A total of 324 individuals, where 292 individuals are from 1000 genomes and 32 individuals are from Simons diversity). This should remove variation in the common ancestor of Sub-Saharan Africans and the Papuans, retaining archaic variants of Neanderthal and Denisova origin as both are present in Papuans, but mainly absent in Africa [7,9]. We also used this filter when analyzing Eurasian populations.

Second we remove variants found in all non Papuan populations (A total of 2751 individuals, where 2504 individuals are from 1000 genomes and 247 individuals are from Simons diversity), only retaining variants that are unique to Papuan populations. This should remove Neanderthal variants that are shared with other non-African populations [1] and also to some extent remove variants of Denisovan origin that are found in Asians and Native Americans [20,21]. Thus removing all variants from the 1000 Genomes Project should enrich for Denisovan segments, while the segments that are found when using Sub-Saharan Africans but not using all 1000 Genomes Project samples as outgroups should be enriched for Neanderthal segments.

We estimated the optimal set of transition and emission parameters for each Papuan individual and found them to be largely consistent across the different datasets, see <u>S4 Fig</u>.

The parameters were converted into estimates of T_{admix} , a, $T_{Ingroup}$ and $T_{Archaic}$ using an average recombination rate of $1.2 \cdot 10^{-8}$ events per base pair per generation and an average mutation rate of $1.25 \cdot 10^{-8}$ mutations per base pair per generation, see Fig 3A and 3B. A table containing all parameters from the model and the corresponding demographic parameters is included in S4 Dataset.

We find that the mean coalescence time between Papuans and non-Papuan individuals is more recent (1,395–1,540 generations ago) than that between Papuans and Sub-Saharan Africans (1,953–2,293 generations ago) reflecting that Papuans are more closely related to other Non-Africans than to Africans. The mean coalescence time between Papuans and other non-



Fig 3. Application of model to Papuan genomes. a) Relationship between modern and archaic humans with the outgroup branches (Sub-Saharan Africans) colored in red. The average coalescence times for ingroup and outgroup $T_{Ingroup}$ and archaic and outgroup $T_{Archaic}$ are shown. The admixture proportions *a* and admixture time T_{admix} are shown for segments that are shared with other non-African populations. b) The outgroup colored in red is now all non-Papuans, and the new demographic parameters are shown. c) The segments that are shared with other Non-

Africans share more variation with the Vindija Neanderthal than they do with the Altai Denisova. Segments that are unique to Papuan individuals share more variation with Altai Denisova than they do with the Vindija Neanderthal. d) Archaic segments that are shared with other non-African populations are shorter than segments that are unique to Papuans (segments with a mean posterior probability > 0.5 are kept).

https://doi.org/10.1371/journal.pgen.1007641.g003

Africans also provides an upper limit for Neanderthal introgression because it happened in their ancestral populations.

Using only Sub-Saharan individuals as an outgroup we find the mean coalescence time between the archaic and outgroup to be between 29,404 and 33,944 generations ago. When using non-Papuans as an outgroup the estimate is between 25,268 and 30,352 generations ago. The lower estimate is likely due to some variants in the common ancestor of Denisovans and Neanderthals having been removed in the latter case.

Using Sub-Saharan Africans as an outgroup we estimate the total admixture proportion of archaic sequence into Papuans to between 4.1–4.4% and the admixture proportion private to Papuans to be 1.5–1.8%. This means that approximately 2.6% is shared with non-Papuans, (Fig 3A).

From the transition parameters, we estimate that the admixture event with non-Africans happened 953–1,254 generations while the Papuan specific admixture event happened 888–1,191 generation ago. Both are likely underestimates as it was for the simulated data with missing data and varying recombination rate. Neanderthal admixture likely occurred closer to 2,000 generations ago after the out of Africa migration [7,22] with Denisovan admixture occurring after that.

We used a threshold of 0.8 posterior probability as it showed a good trade-off between precision and sensitivity on simulated data see Fig 2C and S3 Fig. By comparing to the Vindija Neanderthal [10] and Denisova [2] genomes we find that this cutoff removes around 65% of the segments that don't share variants with any archaic reference genome that were found with a cutoff of 0.5, while only removing 10.4% of the total length of archaic segments, see S5 Fig. Short segments that do not share variants with any archaic reference genome may be enriched for false positive variant calls, or may be deeply coalescing modern human haplotypes, in addition to the possibility of containing material from an unknown archaic source. We note that all segments with very high confidence share variants with Neanderthal and/or Denisova references.

When we use a cutoff of 0.8 we find that 84% of the segments unique to Papuans (80% of the total sequence) shared more variants with the Denisova genome than with the Vindija Neanderthal, and that 78% the segments that are shared with other non-Africans (83% of the total sequence) shared more variants with the Vindija Neanderthal than the Denisova (Fig 3C). This is consistent with most archaic sequence unique to Papuans coming from Denisovans, and most shared archaic sequence coming from Neanderthals.

However, segments that are unique to Papuans are longer on average (94.2 kb) compared to those shared with other non-African populations (76.9 kb), see Fig 3D. The differences in length distributions are not seen as clearly when using Sstar, Sprime or CRF, see S6 Fig. Moreover, the length distributions of archaic segments that are not unique to Papuans (putative Neanderthal segments) are more similar to those found in other non-African populations, see S7 Fig, consistent with a single Neanderthal admixture event.

We compared our archaic segments to those previously reported using other methods [7,9,13]. We find that our method recovers 67% of the archaic sequence found using CRF, 84.9% of the archaic sequence found using Sprime and 74% of the archaic sequence found using Sstar.

Table 1. Amount of sequence of different origins. For different methods and populations, the amounts of sequence (in Mb) are shown in putative archaic segments that share equal numbers of private variants with the Denisova and Vindija Neanderthal (Both), more with Denisova, none with either, or more with Vindija Neanderthal. Neither Sstar nor CRF label segments that do not share variants with the archaic reference genomes. For CRF, segments had to be either more similar to Neanderthal than Denisova or vice versa so they do not report segments that match both equally well. For Sstar the comparison to Denisova was only made for Papuans. Note the Papuans individuals used in Sstar are admixted with East Asians.

Method	Population	Both	Denisova	None	Neanderthal	Total
HMM	Papuan	4.35	83.11	11.54	71.70	170.7
	eastasia	1.48	5.69	9.96	61.37	78.49
	southasia	1.62	5.85	10.12	51.36	68.95
	westeurasia	1.47	2.39	10.14	43.95	57.94
Sstar	Papuan	26.5	43.11	-	49.21	118.82
	eastasia	-	-	-	65.02	65.02
	southasia	-	-	-	55.18	55.18
	westeurasia	-	-	-	51.23	51.23
CRF	Papuan	-	58.17	-	84.72	142.89
	eastasia	-	3.21	-	72.92	76.14
	southasia	-	2.79	-	61.36	64.15
	westeurasia	-	0.68	-	57.29	57.97
Sprime	Papuan	1.04	38.98	13.48	27.85	81.36
	eastasia	0.89	4.29	14.14	60.49	79.81
	southasia	0.76	4.60	15.09	53.83	74.29
	westeurasia	0.76	1.68	14.02	52.22	68.70

https://doi.org/10.1371/journal.pgen.1007641.t001

When comparing the detected segments to the archaic reference genomes our method finds more Denisova segments in Papuans than Neanderthal ones, unlike Sstar and CRF. Our method also detects a smaller amount of additional Denisova segments in East and South East Asians, (Table 1). A dataset with all inferred segments from Papuans and Simons Genome Diversity Project individuals can be found in S5 Dataset.

Discussion

Our method examines the number of private variants in the ingroup compared to the outgroup and implicitly, the distance between the ingroup and the outgroup haplotypes, which interestingly are the features found to carry the highest weights in the new method of Durvasula et al.[14]. Our implementation also allows for missing data, unlike Sprime and Sstar, which may potentially be useful for analysis of ancient DNA samples.

Since emission probabilities are very different between the human and archaic states in our model, we expect a low rate of false positive archaic inference, and this is also what we see in simulations. However, since recombination rates are highly variable, we expect many very short archaic segments and these have a high false negative rate. Our inability to identify these causes us to underestimate the admixture time. Nevertheless, the model does recover the correct size distribution for longer segments (> 50 kb), (S2 Fig). The mean coalescence times of modern and archaic humans are reasonably well estimated in simulations. One issue of interest is that if there were additional super-archaic introgression into the sequenced Denisovan as previously proposed [1], this would cause the mean coalescence time in Denisovan introgressed segments to be greater than that for Neanderthal segments. We did not observe this, although we note that there may be confounding from a low level of Denisovan admixture also present in East Asians which form part of our contrast population, reducing the observed mean divergence.

We report more Denisova segments than approaches relying on the Denisovan reference. This is possibly because our method does not rely on matching to the Altai Denisova sequence, which is believed to be considerably diverged from the source population for "Denisovan" admixture into Papuan ancestors, probably shortly after the split of Denisovans from Neanderthals [1]. Furthermore, because of this early split, many segments may be equally close to the Vindija Neanderthal and the sequenced Denisova sample, and we expect that a fraction of segments introgressed from the Denisovan are more closely related to Vindija and vice versa due to incomplete lineage sorting.

We find no clear evidence for an introgression with a new archaic hominin in Papuans, but we do find segments that do not share variation with any of the sequenced archaic populations. These segments could represent variation in Neanderthals and Denisovans that is not captured by the three high coverage archaic reference genomes, or another source. In the future it will be interesting to compare these segments to other human populations that might also have archaic segments of unknown origin [11,12].

Our model is not restricted to being applied to humans. We have called the admixture source "archaic" so far, which is standard in the human context, but more generally we are modelling a particular form of population structure involving an admixture event from a distantly diverged lineage. We note that other types of population structure, for example involving continual gene flow, could also create signals under our model. Subject to this caveat, the method can be applied where samples have been sequenced from a population that is hypothesized to have received admixture from a perhaps unknown source, and there is comparable data from an outgroup population that did not receive the admixture. The performance of the method depends on the ratio of signal to noise. The signal is stronger the more admixture there is, the more divergent the admixture source is, and the more recently the admixture happened. The noise increases if the outgroup diverged longer ago from the test samples, and if the common ancestor of the ingroup and outgroup had larger population size. The latter problem can be mitigated by sequencing more individuals from the outgroup. Therefore, as an increasing number of individuals are being sequenced in other species, our method could be used to explore introgression in those species, for example chimp and bonobo [23], bears [24], elephants [25] or gibbons [26].

Materials and methods

Simulations

To simulate data we used Msprime [27]. We simulated 5 Papuans and as an outgroup we simulated 500 Africans, 100 Europeans and 100 Asians using demographic parameters from [19]. We simulated data where we varied the recombination rate according to HapMap recombination maps [16] for 5 individuals and removed variants within non-callable regions and variants that were found in the simulated outgroup. We grouped all autosomes into bins of 1000 base pairs and counted the number of variants. For each 1000 bp window we calculated the number of called bases using the repeat masked segments.

We simulated 22 autosomes with varying mutation rate in segments with a mean length of 1 Mb across the genome. The mutation rate in a segment could either be $1.25*10^{-8}$ mutations per base-pair per generation-the average rate, 50% decrease of the average rate or 50% increase of the average rate. We picked the mutation rate in each segment randomly. The choice of segment length and mutation rate is based on [28].

Train parameters and decode segments

We trained and decoded the segments using our HMM, which is available at: <u>https://github.</u> com/LauritsSkov/Introgression-detection/

Data sets

We used 14 Papuans, 71 WestEurasians, 72 East Asians and 39 South Asians individuals from the Simons Genome Diversity Project (SGDP) [18], 40 Papuans from [19] and an additional 35 Papuans [9].

Filtering variants in real data

We used two sets of outgroups. One is all Sub-Saharan Africans (populations: YRI, MSL, ESN) from the 1000 Genomes Project [29] and all Sub-Saharan African populations from SGDP [18] except Masai, Somali, Sharawi and Mozabite, which show signs of out-of-Africa admixture. The other outgroup is all individuals from the 1000 Genomes Project [29] plus all non-Papuans from SGDP. For all human data sets, we also removed sites that fell within repeat-masked [15] regions, and sites that were not in the strict callability mask for the 1000 Genomes Project.

Repeat mask regions

hgdownload.cse.ucsc.edu/goldenpath/hg19/bigZips/chromFaMasked.tar.gz

Strict callability mask for 1000 genomes: ftp://ftp.1000genomes.ebi.ac.uk/vol1/ftp/release/ 20130502/supporting/accessible_genome_masks/StrictMask/

The background mutation rate was calculated using the density of all variants from populations YRI, LWK, GWD, MSL and ESN in windows of 100 Kb divided by the mean variant density of the whole genome.

Comparison to Sstar, Sprime and Conditional Random Field

We called Neanderthal and Denisova segments in the 14 Papuans and compared them to the segments called with CRF with more than 50 posterior probability [7] available at: <u>https://sriramlab.cass.idre.ucla.edu/public/sankararaman.curbio.2016/</u>.

The path to the haplotypes is: <u>summaries/2/denisova/oceania/summaries/haplotypes/</u>CRHOM.thresh-50.length-0.00.haplotypes.

We called Neanderthal and Denisova segments in the 35 Papuans and compared them to the segments called with Sstar with more than 99 posterior probability [9] available at: <u>https://</u>drive.google.com/drive/folders/0B9Pc7_zItMCVWUp6bWtXc2xJVkk

The path to the haplotypes is: introgressed_haplotypes/LL.callsetPNG.mr_0.99.den_calls_ by_hap.bed.merged.by_chr.bed

We called Neanderthal and Denisova segments in the 14 Papuans and compared them to the segments called with Sprime with a score greater than 150,000 [13]. The path to the data is: https://data.mendeley.com/datasets/y7hyt83vxr/1

Supporting information

S1 Fig. Demographic parameters for simulation. The effective population sizes, split times and bottleneck population sizes are shown for the simulated populations. (PDF)

S2 Fig. Total segments and sequence called SIM. The first column show the total number of segments found and the second column show the total amount of sequence that these segments add up to. The rows are different simulation scenarios and the colors of the stacked bar plot show the amount/number of segments that are not found using posterior decoding, where less than half of the segment overlap with the true archaic segments or where more than half of

the segment overlaps with the true archaic segment. (PDF)

S3 Fig. Effect of adjusting cutoff for when to include a putative archaic segment. The amount of sensitivity and precision found when the posterior cutoff is varied. The different colors are different simulation scenarios; ideal is simulated data with constant recombination rate and no missing data, ideal_hap is the same dataset but haploid genomes are used, rec is simulated data where the recombination rate varying along the genome, rec_missing is simulated data with varying recombination rate and missing data and rec_missing_mut is simulated data with varying recombination rate, missing data and varying mutation rate. The admixture proportion for all data is 5%. (PDF)

S4 Fig. Parameter estimation of Papuans. The different subpanels show the estimates for the parameters t_admix, a, T_ingroup and T_archaic depending on which outgroup was used (Sub-Saharan Africans) or the whole world (non-Papuans). There is a separate bar for each individual, and the bars are colored according to which dataset they came from. (PDF)

S5 Fig. Segment distributions as a function of posterior probability. Distributions of the number (left) and total length (right) of segments with mean posterior probability as on the x axis. Numbers are given for all 89 Papuans, called with Sub-Saharan Africans as the outgroup, and with a threshold of 0.5.

(PDF)

S6 Fig. Length distribution of inferred segments for other methods. The length distribution of all Denisova and Neanderthal segments found using conditional random field (CRF), the hidden Markov model (HMM) and Sstar. For our HMM, Neanderthal are those segments that are shared with other non-African populations and Denisova are those unique to Papuans. (PDF)

S7 Fig. Length distribution of Asians, Europeans and Papuans. The length distributions of segments unique to Papuans (Denisova) and segments shared with other non-African populations (Neanderthal) are shown for segments found using four different population groups. (PDF)

S1 Dataset. Converting model parameters to demographic parameters for haploid and diploid data.

(PDF)

S2 Dataset. Python script to simulate data. The python script is using the Msprime package to simulate whole genome sequences under a given scenario. (PY)

S3 Dataset. This spreadsheet contains two tabs. In the tab "Simulated data parameters" a table with each of the 15 simulated individuals. For each individual the parameter estimates from the model and the corresponding demographic parameter is shown. In the tab "Simulated data precision_sens" the precision and sensitivity are shown for each individual with different cutoffs.

(XLSX)

S4 Dataset. This spreadsheet contains a table with each of the 89 Papuans and the 182 individuals from Simons diversity project. For each individual the parameter estimates from the model and the corresponding demographic parameter is shown. (XLSX)

S5 Dataset. This table contains the inferred archaic segments for the 89 Papuans and the 182 individuals from Simons diversity project. The columns are shown in the header. (TXT)

Author Contributions

Conceptualization: Laurits Skov, Vladimir Shchur, Richard Durbin.

Data curation: Ruoyun Hui.

Formal analysis: Laurits Skov, Ruoyun Hui.

Funding acquisition: Mikkel Heide Schierup, Richard Durbin.

Investigation: Laurits Skov.

Methodology: Laurits Skov.

Resources: Laurits Skov.

Software: Laurits Skov.

Supervision: Richard Durbin.

Validation: Laurits Skov.

Visualization: Laurits Skov.

- Writing original draft: Laurits Skov, Ruoyun Hui, Asger Hobolth, Aylwyn Scally, Mikkel Heide Schierup, Richard Durbin.
- Writing review & editing: Laurits Skov, Ruoyun Hui, Vladimir Shchur, Asger Hobolth, Aylwyn Scally, Mikkel Heide Schierup, Richard Durbin.

References

- Prufer K, Racimo F, Patterson N, Jay F, Sankararaman S, et al. (2014) The complete genome sequence of a Neanderthal from the Altai Mountains. Nature 505: 43–49. <u>https://doi.org/10.1038/nature12886</u> PMID: 24352235
- Meyer M, Kircher M, Gansauge MT, Li H, Racimo F, et al. (2012) A high-coverage genome sequence from an archaic Denisovan individual. Science 338: 222–226. https://doi.org/10.1126/science.1224344 PMID: 22936568
- Huerta-Sanchez E, Jin X, Asan Bianba Z, Peter BM, et al. (2014) Altitude adaptation in Tibetans caused by introgression of Denisovan-like DNA. Nature 512: 194–197. <u>https://doi.org/10.1038/nature13408</u> PMID: 25043035
- Racimo F, Gokhman D, Fumagalli M, Ko A, Hansen T, et al. (2017) Archaic Adaptive Introgression in TBX15/WARS2. Mol Biol Evol 34: 509–524. https://doi.org/10.1093/molbev/msw283 PMID: 28007980
- 5. Dannemann M, Prufer K, Kelso J (2017) Functional implications of Neandertal introgression in modern humans. Genome Biol 18: 61. https://doi.org/10.1186/s13059-017-1181-7 PMID: 28366169
- Seguin-Orlando A, Korneliussen TS, Sikora M, Malaspinas AS, Manica A, et al. (2014) Paleogenomics. Genomic structure in Europeans dating back at least 36,200 years. Science 346: 1113–1118. https://doi.org/10.1126/science.aaa0114 PMID: 25378462
- Sankararaman S, Mallick S, Patterson N, Reich D (2016) The Combined Landscape of Denisovan and Neanderthal Ancestry in Present-Day Humans. Curr Biol 26: 1241–1247. https://doi.org/10.1016/j.cub. 2016.03.037 PMID: 27032491
- Plagnol V, Wall JD (2006) Possible ancestral structure in human populations. PLoS Genet 2: e105. https://doi.org/10.1371/journal.pgen.0020105 PMID: 16895447

- Vernot B, Tucci S, Kelso J, Schraiber JG, Wolf AB, et al. (2016) Excavating Neandertal and Denisovan DNA from the genomes of Melanesian individuals. Science 352: 235–239. <u>https://doi.org/10.1126/ science.aad9416 PMID: 26989198</u>
- 10. Prufer K, de Filippo C, Grote S, Mafessoni F, Korlevic P, et al. (2017) A high-coverage Neandertal genome from Vindija Cave in Croatia. Science.
- Hsieh P, Woerner AE, Wall JD, Lachance J, Tishkoff SA, et al. (2016) Model-based analyses of wholegenome data reveal a complex evolutionary history involving archaic introgression in Central African Pygmies. Genome Res 26: 291–300. https://doi.org/10.1101/gr.196634.115 PMID: 26888264
- Mondal M, Casals F, Xu T, Dall'Olio GM, Pybus M, et al. (2016) Genomic analysis of Andamanese provides insights into ancient human migration into Asia and adaptation. Nat Genet 48: 1066–1070. https://doi.org/10.1038/ng.3621 PMID: 27455350
- Browning SR, Browning BL, Zhou Y, Tucci S, Akey JM (2018) Analysis of Human Sequence Data Reveals Two Pulses of Archaic Denisovan Admixture. Cell 173: 53–61 e59. <u>https://doi.org/10.1016/j.</u> cell.2018.02.031 PMID: 29551270
- 14. Durvasula A, Sankararaman S (2018) Recovering signals of ghost archaic admixture in the genomes of present-day Africans. bioRxiv.
- 15. Smit AFA, Hubley R, Green P (2013) RepeatMasker Open 4.0. RepeatMasker Open 40.
- International HapMap C, Frazer KA, Ballinger DG, Cox DR, Hinds DA, et al. (2007) A second generation human haplotype map of over 3.1 million SNPs. Nature 449: 851–861. https://doi.org/10.1038/ nature06258 PMID: 17943122
- Vernot B, Akey JM (2014) Resurrecting surviving Neandertal lineages from modern human genomes. Science 343: 1017–1021. https://doi.org/10.1126/science.1245938 PMID: 24476670
- Mallick S, Li H, Lipson M, Mathieson I, Gymrek M, et al. (2016) The Simons Genome Diversity Project: 300 genomes from 142 diverse populations. Nature 538: 201–206. <u>https://doi.org/10.1038/</u> nature18964 PMID: 27654912
- Malaspinas AS, Westaway MC, Muller C, Sousa VC, Lao O, et al. (2016) A genomic history of Aboriginal Australia. Nature 538: 207–214. https://doi.org/10.1038/nature18299 PMID: 27654914
- Skoglund P, Jakobsson M (2011) Archaic human ancestry in East Asia. Proc Natl Acad Sci U S A 108: 18301–18306. https://doi.org/10.1073/pnas.1108181108 PMID: 22042846
- 21. Qin P, Stoneking M (2015) Denisovan Ancestry in East Eurasian and Native American Populations. Mol Biol Evol 32: 2665–2674. https://doi.org/10.1093/molbev/msv141 PMID: 26104010
- 22. Fu Q, Li H, Moorjani P, Jay F, Slepchenko SM, et al. (2014) Genome sequence of a 45,000-year-old modern human from western Siberia. Nature 514: 445–449. <u>https://doi.org/10.1038/nature13810</u> PMID: 25341783
- de Manuel M, Kuhlwilm M, Frandsen P, Sousa VC, Desai T, et al. (2016) Chimpanzee genomic diversity reveals ancient admixture with bonobos. Science 354: 477–481. <u>https://doi.org/10.1126/science.</u> aag2602 PMID: 27789843
- 24. Liu SP, Lorenzen ED, Fumagalli M, Li B, Harris K, et al. (2014) Population Genomics Reveal Recent Speciation and Rapid Evolutionary Adaptation in Polar Bears. Cell 157: 785–794. https://doi.org/10. 1016/j.cell.2014.03.054 PMID: 24813606
- Palkopoulou E, Lipson M, Mallick S, Nielsen S, Rohland N, et al. (2018) A comprehensive genomic history of extinct and living elephants. Proc Natl Acad Sci U S A 115: E2566–E2574. https://doi.org/10. 1073/pnas.1720554115 PMID: 29483247
- Carbone L, Harris RA, Gnerre S, Veeramah KR, Lorente-Galdos B, et al. (2014) Gibbon genome and the fast karyotype evolution of small apes. Nature 513: 195–201. https://doi.org/10.1038/nature13679 PMID: 25209798
- Kelleher J, Etheridge AM, McVean G (2016) Efficient Coalescent Simulation and Genealogical Analysis for Large Sample Sizes. PLoS Comput Biol 12: e1004842. https://doi.org/10.1371/journal.pcbi. 1004842 PMID: 27145223
- Smith TCA, Arndt PF, Eyre-Walker A (2018) Large scale variation in the rate of germ-line de novo mutation, base composition, divergence and diversity in humans. PLoS Genet 14: e1007254. <u>https://doi.org/ 10.1371/journal.pgen.1007254 PMID: 29590096</u>
- Genomes Project C, Auton A, Brooks LD, Durbin RM, Garrison EP, et al. (2015) A global reference for human genetic variation. Nature 526: 68–74. https://doi.org/10.1038/nature15393 PMID: 26432245